Author's response to reviews

Title: Viral pathogens in children hospitalized with features of central nervous system infection in a malaria-endemic region of Papua New Guinea

Authors:

Moses Laman (drmlaman@yahoo.com)
Ilomo Hwaiwhanje (wohuierekei@hotmail.com)
Cathy Bona (cathytiribona@gmail.com)
Jonathan Warrell (jonathan.warrel@pngimr.org.pg)
Susan Aipit (susan.aipit@yahoo.com)
David Smith (david.smith@health.wa.gov.au)
Joanna Noronha (Joanna.noronha@health.wa.gov.au)
Peter Siba (peter.siba@pngimr.org.pg)
Ivo Mueller (ivomueller@fastmail.fm)
Inoni Betuela (ibetuela@gmail.com)
Timothy ME Davis (tim.davis@uwa.edu.au)
Laurens Manning (laurens.manning@uwa.edu.au)

Version: 2
Date: 2 September 2014

Author’s response to reviews:

Line and page numbers added. Page breaks deleted.